Title: Adequate sleep among adolescents is positively associated with health status and health-related behaviors

Authors:
    Mei-Yen Chen (meiyen@mail.cgit.edu.tw)
    Edward K Wang (mymcwang@yahoo.com.tw)
    Yi-Jong Jeng (yijeng@mail.cgit.edu.tw)

Version: 5 Date: 25 February 2006

Author's response to reviews: see over
To editor,

Thank you for giving us the convenient resource of editing company. We’ve made the suggested revisions and invited an editor who is a native speaker of English to edit the manuscript.

1. Running title: We’ve removed the running title from the title page.
2. Keywords section: We’ve removed the keywords section from the title page.
3. We’ve removed any tracking from the manuscript text.
4. We’ve renamed the section as “Methods”.
5. We’ve provided a statement about Dr. Jeng’s contribution in this manuscript.
6. We’ve reformatted the tables in black and white.
7. We’ve also followed the rule of BMC-Author’s checklist format.

Responses to Reviewer 1’s comments:

1. We’ve asked a native English speaker (from Biomedes-Manuscript Presentation Service from UK) to revise our manuscript. Thank you.